Author's response to reviews

Title: Neurobehavioral effects of transportation noise in primary school children: A cross-sectional study.

Authors:

Elise van Kempen (Elise.van.Kempen@rivm.nl)
Irene van Kamp (Irene.van.Kamp@rivm.nl)
Erik Lebret (Erik.Lebret@rivm.nl)
Jan Lammers (Jan.Lammers@tno.nl)
Harry Emmen (hemmen@worldonline.nl)
Stephen Stansfeld (S.A.Stansfeld@qmul.ac.uk)

Version: 4 Date: 29 April 2010

Author's response to reviews:

Dear Sir, dear Madam,

Further to your letter of 23th April 2010, please find enclosed a new revised version of the manuscript for resubmission to Environmental Health. The manuscript has been revised to take into account the comments of the Environmental Health Editorial Team. Each of the comments is addressed in below. We hope that you will reconsider our manuscript for publication.

On the title page, the email addresses should be listed with Email addresses as heading and underneath the initials of each author followed by the email address separating the initials and addresses with a colon.

As requested, we added a heading ‘Email addresses’ and underneath we included the initials of each author followed by their email address. The initials and email addresses were separated with a colon.

The periods after the headings in the abstract and elsewhere in the text should be removed.

This is done as requested.

The square brackets in the first reference should be removed.

We removed the square brackets.

The information about the additional files on page 34 should be removed.

This is done as requested.

Finally, it looks like the Abbreviations section needs to be the other way round. Please list first the abbreviation, then the explanation.

We have checked the Abbreviations section and indeed something went wrong: we forgot the first abbreviation and the rest appeared to be the other way round.
As requested, we revised this section.

With kind regards,
Elise van Kempen, PhD
Researcher